Author’s response to reviews

Title: Using routine clinical and administrative data to produce a dataset of attendances at Emergency Departments following self-harm

Authors:

Catherine Polling Dr (catherine.polling@kcl.ac.uk)
Alex Tulloch Dr (alex.tulloch@kcl.ac.uk)
Shikha Banerjee Miss (shikha.banerjee@kcl.ac.uk)
Sean Cross Dr (sean.cross@slam.nhs.uk)
Rina Dutta Dr (rina.dutta@kcl.ac.uk)
Paul Dargan Dr (paul.dargan@gstt.nhs.uk)
David Wood Dr (david.wood@gstt.nhs.uk)
Matthew Hotopf Prof (matthew.hotopf@kcl.ac.uk)

Version: 5  Date: 23 June 2015

Author’s response to reviews:

Thank you for considering our manuscript, “Using routine clinical and administrative data to produce a dataset of attendances at Emergency Departments following self-harm” by C. Polling, A. Tulloch, S. Banerjee, R Dutta, S Cross, P Dargan, D. Wood and M Hotopf.

Please find enclosed our revised manuscript. The revisions address editorial requests as outlined below.

• Authors’ contact details have been moved to the first page of the manuscript
• The contribution of Rina Dutta has been clarified in the Authors’ contribution section
• The ‘Ethical and clinical governance approvals’ section has been removed from the end and its text incorporated into the text of the manuscript in the Methods section.
• The List of abbreviations has been moved to after Conclusions

Yours sincerely,

Dr Catherine Polling

Corresponding author